Author's response to reviews

Title: The Ilizarov external fixator - a useful alternative for the treatment of proximal tibial fractures A prospective observational study of 30 consecutive patients

Authors:

Telmo Ramos (tramos@tele2.se)
Carl Ekholm (carl.ekholm@vgregion.se)
Bengt Eriksson (bengt.eriksson@vgregion.se)
Jón Karlsson (jon.karlsson@vgregion.se)
Lars Nistor (lars.nistor@vgregion.se)

Version: 2 Date: 16 March 2012

Author's response to reviews:

waiting for your revisions